Author's response to reviews

Title: The Effectiveness of Fermented Turmeric Powder in Subjects with Elevated Alanine Transaminase Levels: a randomised controlled study

Authors:

Sang-Wook Kim (clickm@chonbuk.ac.kr)
Ki-Chan Ha (omphalos@chonbuk.ac.kr)
Eun-Kyung Choi (ekchoi@ibctc.org)
Min-Gul Kim (mgkim@ibctc.org)
Dae-Young Kwon (dykwon@kfri.re.kr)
Hye-Jeong Yang (yhi@kfri.re.kr)
Min-Jeong Kim (well@kfri.re.kr)
Hee-Joo Kang (anni0216@hanmail.net)
Hyang-Im Back (hyangim100@gmail.com)
Sun-Young Kim (sykim@ibctc.org)
Soo-Hyun Park (shpark@ibctc.org)
Hum-Young Baek (bhy0323@paran.com)
Yong-Jae Kim (yids06@nate.com)
Joon-Yeol Lee (jylee94@nate.com)
Soo-Wan Chae (soowan@jbnu.ac.kr)

Version: 5 Date: 28 December 2012

Author's response to reviews: see over
Cover letter

Dear Editors,

My MS ID is 1006270834773725. I was revised my manuscript as pointed out previous.

Reviewer: Song-Yi Kim

1. Is the question posed by the authors well defined? The safety endpoint such as results of serum creatinine was not mentioned.---Revised
2. Are the methods appropriate and well described?---Revised
3. Are the data sound? ----Revised
4. Are the discussion and conclusions well balanced and adequately supported by the data?---Revised
5. Are limitations of the work clearly stated?—Revised
6. Do the title and abstract accurately convey what has been found?---Revised
7. Is the writing acceptable?---- I wish to apply for English editing services for my revised manuscript
8. Minor comments---Revised

Reviewer: Yi-Hung Chen

1. When sample size calculation was performed, why 10.2 IU/L was considered a clinically relevant difference between the two groups?---Revised
2. What is the component of placebo capsule? Revised
3. Is there any effort made to keep the FTP quality consistent during the 12-week?---Revised